Author's response to reviews

Title: Educational Potential of a Virtual Patient System for Caring for Traumatized Patients in Primary Care

Authors:

Solvig Ekblad (Solvig.Ekblad@ki.se)
Richard F Mollica (RMOLLICA@PARTNERS.ORG)
Uno Fors (uno@dsv.su.se)
Ioannis Pantziaras (Ioannis.Pantziaras@ki.se)
James Lavelle (JLAVELLE@PARTNERS.ORG)

Version: 2 Date: 21 February 2013

Author's response to reviews:

We have revised the title in the covering letter according to the mail from Aldrin Joseph Ulep Feb 18.